Author's response to reviews

Title: Detection of mammaglobin mRNA in peripheral blood is associated with high grade breast cancer: interim results of a prospective cohort study

Authors:

Kaidi Mikhitarian (mikhitariank@health.missouri.edu)
Renee Hebert Martin (hebertrl@musc.edu)
Megan Baker Ruppel (bakermk@musc.edu)
William E Gillanders (gillandersw@wudosis.wustl.edu)
Rana Hoda (hodar@musc.edu)
Del H Schutte (schutted@musc.edu)
Kathi Callahan (callahanka@musc.edu)
Michael Mitas (mitasm@musc.edu)
David J Cole (coledj@musc.edu)

Version: 5 Date: 25 January 2008

Author's response to reviews:

January 25, 2008

Manuscript #: 1111933707149567

Title: Detection of mammaglobin mRNA in peripheral blood is associated with high grade breast cancer: interim results of a prospective cohort study

Responses to the formatting changes requested:

Country - Please make sure you include the country, and format it in sentence case, for example: India (not INDIA).
Please change: Materials and Methods -> Methods
List of abbreviations -> Abbreviations
Please change FIGURE and TABLE to Figures and Tables.
Please remove the current address affiliations, include them as normal i.e. give them a number and link them in the same way as affiliation 1.
Formatting changes has been made as requested.

Endnote/Refman - There is a discrepancy between the number of references in your manuscript reference list, and the number extracted by our system.
References have been reformatted.

Tables - Tables must be divided into cells / fields and be oriented in a portrait
layout, not landscape. The borders of tables must be visible black lines. Tables generated with tabbed text or included as an embedded image are not acceptable, nor are bullet points. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1.

Tables have been reformatted using MS Word auto-format template: simple 1.

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. The figure has been cropped and submitted as a PowerPoint slide.

References - The reference list should contain all authors’ names, if there are 30 or below, the term ‘et al.’ should not be used. If there are more than 30 names a minimum of 30 should be listed, and then et al can be used.

References were formatted using BioMed Central style option in EndNote X. This format style shows maximum of 10 authors and not 30 authors. Please let me know if this is a problem. If yes, please instruct how to change it.

Please do not hesitate to contact me if there are any additional concerns.

Sincerely,

Kaidi Mikhitarian, M.D.